Author's response to reviews

Title: Village doctor-assisted case management of rural patients with schizophrenia: A randomized control study

Authors:

Wenjie Gong (gongwenjie@csu.edu.cn)
Dong Xu (romanxu@uw.edu)
Liang Zhou (zhouliang1976@gmail.com)
Henry S Brown III (Henry.S.Brown@uth.tmc.edu)
Kirk L Smith (klsmith@utmb.edu)
Shuiyuan Xiao (xiaosy@csu.edu.cn)

Version: 2 Date: 26 October 2013

Author's response to reviews:

As discussed with the editor, the "result" section is not necessary for a study protocol. So we only made a litter format and words revision according to the requirement in your letter.

1. Added a title page as page1, including the title, authors, author affiliations and corresponding author details.
2. Canceled the figures and table from the main manuscript document, they will be uploaded as separate files via the online submission system.
3. Exchanged the order of "Trial status" and "Discussion" sections in accordance with the Instructions guide. And added the information on the stage of the development of the pilot study in the trial status section.
4. Corrected some spelling and grammar errors.
5. Sent the funding letter and approval document of the ethics committee for our study as the email attachments to editor Aiza.